Author's response to reviews

Title: Maternal deaths in eastern Indonesia: twenty years and still walking, an ethnographic study

Authors:

Suzanne Belton (suzanne.belton@menzies.edu.au)
Bronwyn Myers (bronwyn.myers@cdu.edu.au)

Version: 2 Date: 25 June 2013

Author's response to reviews: see over
Dear Editor

Please note that I have made your suggested changes to this manuscript.

1. Title page: done
2. Competing interests: done
3. Financial competing interests : done
4. Authors' information: done
5. Acknowledgements: done
6. Ethics committee: done

I have also used BMC referencing.

Thank you

Suzanne Belton